Author's response to reviews

Title: Low CD4 Count plus Coma Predicts Cryptococcal Meningitis in Tanzania

Authors:

- Peter R Kisenge (pkisenge@yahoo.com)
- Alexander T Hawkins (ath3a@virginia.edu)
- Venance P Maro (venmaro@yahoo.co.uk)
- John PD Mchele (johnmchele@yahoo.com)
- Ndelilia S Swai (lilia2005tz@yahoo.co.uk)
- Andreas Muller (mueller.moshi@web.de)
- Eric R Houpt (erh6k@virginia.edu)

Version: 5 Date: 27 April 2007

Author's response to reviews:

Dear Ms. Makri/BMC Infectious Diseases Editor,

Thank you for notice regarding the publication of 534883921173823 "Low CD4 Count plus Coma Predicts Cryptococcal Meningitis in Tanzania" for publication. We have made all minor editorial changes requested.

Kind regards,

Eric R Houpt MD
University of Virginia
Corresponding Author